

Ciosi, M., Masiga, D.K., and Turner, C.M. (2014) Laboratory colonisation and genetic bottlenecks in the tsetse fly Glossina pallidipes. PLoS Neglected Tropical Diseases, 8 (2). e2697. ISSN 1935-2727

Copyright © 2014 The Authors.

http://eprints.gla.ac.uk/89385/

Deposited on: 14 February 2014

Enlighten – Research publications by members of the University of Glasgow http://eprints.gla.ac.uk

Laboratory Colonisation and Genetic Bottlenecks in the Tsetse Fly *Glossina pallidipes*

Marc Ciosi^{1,2}*, Daniel K. Masiga^{1,2}*, Charles M. R. Turner²*

1 Molecular Biology and Bioinformatics Unit, ICIPE, Nairobi, Kenya, 2 Institute of Infection, Immunity & Inflammation, University of Glasgow, Glasgow, United Kingdom

Abstract

Background: The IAEA colony is the only one available for mass rearing of *Glossina pallidipes*, a vector of human and animal African trypanosomiasis in eastern Africa. This colony is the source for Sterile Insect Technique (SIT) programs in East Africa. The source population of this colony is unclear and its genetic diversity has not previously been evaluated and compared to field populations.

Methodology/Principal Findings: We examined the genetic variation within and between the IAEA colony and its potential source populations in north Zimbabwe and the Kenya/Uganda border at 9 microsatellites loci to retrace the demographic history of the IAEA colony. We performed classical population genetics analyses and also combined historical and genetic data in a quantitative analysis using Approximate Bayesian Computation (ABC). There is no evidence of introgression from the north Zimbabwean population into the IAEA colony. Moreover, the ABC analyses revealed that the foundation and establishment of the colony was associated with a genetic bottleneck that has resulted in a loss of 35.7% of alleles and 54% of expected heterozygosity compared to its source population. Also, we show that tsetse control carried out in the 1990's is likely reduced the effective population size of the Kenya/Uganda border population.

Conclusions/Significance: All the analyses indicate that the area of origin of the IAEA colony is the Kenya/Uganda border and that a genetic bottleneck was associated with the foundation and establishment of the colony. Genetic diversity associated with traits that are important for SIT may potentially have been lost during this genetic bottleneck which could lead to a suboptimal competitiveness of the colony males in the field. The genetic diversity of the colony is lower than that of field populations and so, studies using colony flies should be interpreted with caution when drawing general conclusions about *G. pallidipes* biology.

Citation: Ciosi M, Masiga DK, Turner CMR (2014) Laboratory Colonisation and Genetic Bottlenecks in the Tsetse Fly *Glossina pallidipes*. PLoS Negl Trop Dis 8(2): e2697. doi:10.1371/journal.pntd.0002697

Editor: Jesus G. Valenzuela, National Institute of Allergy and Infectious Diseases, United States of America

Received October 26, 2013; Accepted December 31, 2013; Published February 13, 2014

Copyright: © 2014 Ciosi et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: We thank the Wellcome Trust for financial support (grant number: 093692). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: marc.ciosi@glasgow.ac.uk

9 These authors contributed equally to this work.

Introduction

Tsetse flies are vectors of African trypanosomes, parasites that cause human and animal African trypanosomiases. Tsetse control is one of the main methods used to combat the disease [1]. To be efficient, tsetse control attempts should involve a combination of methods that are efficient at both high and low population densities [2]. The sterile insect technique (SIT) is particularly important for control as it is one of the few control methods that is efficient at low densities [2]. SIT is effected by the sequential mass release of sterile males generated from a laboratory colony into an infested area. When the sterile males outnumber the wild males they mate more successfully with wild females that will then produce no offspring. SIT has proven to be an efficient control technique in different parts of the world mainly in pests of crops but was also successfully used to eradicate the New World screwworm Cochliomyia hominivorax, a veterinary pest, in North and Central America [reviewed in 3].

To be successful, SIT programmes need to overcome a number of potential genetic difficulties. Firstly, genetic and phenotypic differentiation can cause mating barriers between wild populations which can make SIT less effective depending on the geographical origin of the sterile insects [4]. Second, the establishment of a laboratory colony for SIT is likely to be associated with strong selection pressure toward laboratory adaptation and loss of genetic diversity compared to the field population of origin. Third, as SIT often involves the release of males, the use of an imbalanced sex ratio within the colony (a common practice) is likely to increase genetic drift in the colony and thus loss of genetic diversity compared with the original field population. Laboratory adaptation and genetic diversity loss can be associated with a loss of field competitiveness [4,5] which can limit the efficiency of SIT.

SIT has proven useful in tsetse control, the most striking example being the eradication of *Glossina austeni* from Unguja Island (Zanzibar), Tanzania through an area-wide integrated tsetse eradication project terminated by a phase of SIT [6]. Following this success, new tsetse SIT programs have started including one targeting *Glossina pallidipes*, a vector of human African trypanosomiasis [7–9] and arguably the main vector of animal African

Author Summary

There is only one mass reared laboratory colony of Glossina pallidipes, a vector of human African trypanosomiasis and arguably the main vector of animal African trypanosomiasis in eastern Africa. This colony is the main one used for basic research on this species and is intended to be used for Sterile Insect Technique (SIT) programs for control of field populations. The origins of this colony are not clear and the extent to which it is genetically representative of the species is unknown. Using population genetics analyses to compare the current colony with two potential source populations we have shown that the colony is from the Kenya/Uganda border and that its foundation and establishment in the laboratory were associated with a genetic bottleneck, i.e. reduction of genetic variation due to increased genetic drift in a population of reduced size. As a consequence, the genetic diversity of the colony is lower than that of G. pallidipes field populations.

trypanosomiasis in eastern Africa [10]. Pilot releases of sterile G. *pallidipes* males are about to be performed [11].

The IAEA G. pallidipes colony is one of the very few laboratory colonies of this species worldwide and the only one with which mass rearing has been achieved to date. This is the reason why it has been used to start mass rearing in all the SIT Facilities that are currently rearing G. pallidipes [12,13]. The IAEA G. pallidipes colony is generally considered to have been established from the laboratory colony of the University of Amsterdam, the Netherlands which was in turn established from wild pupae collected from Lugala, Uganda in 1975 [14]. Several details associated with the origin of the colony are unclear however. For example, the 'start date' for the IAEA colony varies between publications [see 15,16-18,19 for different dates assumed for the initiation of the IAEA colony]. Moreover, an IAEA report from March 1987 indicate that "A new colony of G. pallidipes was initiated from pupae kindly donated by the Tsetse Research Laboratory, Bristol, England." and this is believed to have been the start of the current IAEA colony [12,17,20]. This potential other origin is important because the G. pallidipes sent to IAEA in 1987 may have originated from northern Zimbabwe [21]. In summary, published accounts do not allow the identification of a single field population as the source of the current G. pallidipes IAEA colony and cannot rule out the possibility of admixture between colonies of Zimbabwean and Ugandan origin. This lack of clarity is potentially important because of the high genetic differentiation between G. pallidipes populations from Uganda and Zimbabwe [22]. If the IAEA colony was the result of an admixture between those populations it would harbour much of the genetic diversity present in the species. This could be an advantage for future SIT programs as it could limit potential mating barriers with field populations targeted by SIT control.

Molecular markers and population genetics can be used to reconstruct the demographic history of populations, thereby providing insight into aspects of population establishment and colonisation such as founder effect, population bottleneck and demographic expansion [e.g. 23,24–26]. In that context, the recently developed Approximate Bayesian Computation (ABC) [27–29] has proven useful to identify complex and unexpected colonisation histories [30,31]. ABC allows the quantitative comparison of complex demographic or evolutionary scenarios and the estimation of parameters of interest based on molecular and historical data. It has been shown to be a powerful method to compare complex demographic and phylogeographic scenarios based on population genetics data sets [e.g. 23,30,32–34]. We examined here the genetic variation within and between the IAEA colony and its potential source populations in north Zimbabwe and the Kenya/Uganda border to retrace the demographic history of the IAEA colony. We performed classical population genetics analyses and also combined historical information and microsatellite data in a quantitative analysis of the genetic variation using ABC methods. We addressed the following specific questions: (i) Does the IAEA colony result from an admixture between the two potential source populations or originate from a single source? (ii) What is the level of genetic drift associated with laboratory colonisation and the rearing of the colony between its foundation and the present?

Materials and Methods

Sample collection and DNA extraction

G. pallidipes were obtained from the IAEA Seibersdorf colony and from its potential source populations (Rukomeshi, Zimbabwe and the Uganda/Kenya border, Table 1). Samples of the IAEA colony obtained in 2012 (30 flies) and 2013 (26 flies) were used to better represent the contemporary demographic stochasticity within the colony. The sample (31 flies) used to represent the Uganda/Kenya border population was collected in 2000 in Kapesur near Busia, Kenya. This location is adequate as there is no genetic differentiation between sites within this area [35]. The Rukomeshi sample (34 flies), collected in 2006, was previously analysed for the presence and genetic diversity of the salivary gland hypertrophy virus by Kariithi et al [36]. Flies sampled in Busia were dried after collection and stored in 95% ethanol. IAEA colony specimens were stored in 95% EtOH. For Busia and IAEA colony samples, DNA extractions were carried out from the abdomen of each fly using the Qiagen DNeasy blood and tissue kit following the manufacturer indications and using a final elution of 100 µl. DNA extractions from Rukomeshi sample were carried out as indicated in Kariithi et al [36]

Distinguishing between *G. pallidipes* and *G. fuscipes* can prove challenging when specimens have been stored in 95% EtOH because of the alteration of colours that are the main characteristics that allow distinguish between these two species [37]. To make sure no confusion was made at sampling between *G. pallidipes* and *G. fuscipes fuscipes* in Busia (where these species are sympatric), we tested whether the microsatellite locus GmmF10 could be amplified in flies collected in Busia as this marker can be amplified in *G. pallidipes* but not in *G. fuscipes* [38,39].

Microsatellite genotyping

15 polymorphic microsatellites loci previously described in the literature [38-42] were evaluated for their use in population genetics analyses of G. pallidipes using multiplex PCR as described in Supplementary file S1. Loci were combined into multiplex reactions with the help of Multiplex Manager v1.2 [43] in an analysis of 2 millions iterations, a primer complementarity threshold of 7 and a minimum distance between loci of the same dye color of 26 bp. The multiplex reactions were fine tuned by hand. After a validation step (fully described in Supplementary file S1) we ended up using 9 microsatellites loci in two multiplex PCR. Multiplex reaction α contained loci GmmK06, GmmC17, GpC10b, GpC101, GpB115, GpCAG133. Multiplex reaction β contained loci: GmmA06, GpA19a and GpC26 (Table S1). This resulted in a primer complementary threshold of 6 within multiplex reactions and of a minimum distance between loci of the same dye colour of 58 bp. Multiplex PCR were carried out in a total volume of 10 µl containing 2 µl of template DNA solution, 1X Qiagen Multiplex PCR mix and 0.2 µM of each primers

Table 1. G. pallidipes samples used in this study plus HWE tests.						
Area/country	Sites/colony	Collection date	Latitude(°N), Longitude(°E)	Sample size	HWE	F _{IS}
North Zimbabwe	Rukomeshi	2006	-16.13, 29.40	34	<0.01	0.126
Uganda/Kenya border	Busia	April 2000	0.61, 34.30	31	0.906	-0.025
Seibersdorf, Austria	IAEA	March 2012	-	30	0.997	-0.090
		April 2013	-	26	0.999	0.012
North Zimbabwe Uganda/Kenya border Seibersdorf, Austria	Rukomeshi Busia IAEA	2006 April 2000 March 2012 April 2013	- 16.13, 29.40 0.61, 34.30 - -	34 31 30 26	<0.01 0.906 0.997 0.999	0.126 -0.025 -0.090 0.012

Table 1. *G. pallidipes* samples used in this study plus HWE tests.

Sample size: number of individuals genotyped per sample. HWE: p-value of the test for deviation from Hardy–Weinberg equilibrium. doi:10.1371/journal.pntd.0002697.t001

except for locus GpC10b (0.3 µM of each primers). Forward PCR primers 5' labelled with a fluorescent dye were used to allow the PCR products to be detected on an automated DNA sequencer. The PCR cycling conditions for both multiplex PCRs were (95°C, 15 min); 25 cycles of (94°C, 30 s), (55°C, 90 s) and (72°C, 60 s); $(60^{\circ}C, 30 \text{ min})$. 1 µl of a 1/20 or 1/30 dilution of the multiplex PCR products were analysed by electrophoresis in combination with the GeneScan-500 LIZ size standard (Applied Biosystems) by DNA Sequencing & Services (MRCPPU, College of Life Sciences, University of Dundee, Scotland, www.dnaseq.co.uk) using Applied Biosystems Big-Dye Ver 3.1 chemistry on an Applied Biosystems model 3730 automated capillary DNA sequencer. The size estimation of amplified microsatellites was performed using GENEMARKER v2.2.0 (SoftGenetics). The Excel Macro Autobin v0.9 [44] was then used on the raw data set of amplified microsatellites sizes to automatically detect relevant gaps in size and help delimit allele "bins" (Table S2). The allele "bins" defined using AUTOBIN were then used within GENEMARKER to automatically bin the alleles. Each peak was then checked manually.

Microsatellite data are available from the Dryad Digital Repository: http://dx.doi.org/10.5061/dryad.bt612

Genetic variation within samples

Genetic variation within samples was assessed using the mean number of alleles per locus (Na) and the mean expected heterozygosity (H) [45] computed with GENECLASS 2 ver. 2.0.h [46]. The coefficient of inbreeding F_{IS} was estimated with GENEPOP ON THE WEB [47,48]. For comparisons of Na values between samples, allelic richness (AR) was estimated on the basis of minimum sample size with FSTAT 2.9.3.2 [49]. The significance of differences in AR and H between samples was assessed with the nonparametric Friedman and Wilcoxon sign rank tests (with the locus as a repetition unit). Deviation from Hardy–Weinberg equilibrium (HWE) was assessed with the probability test approach, using GENEPOP ON THE WEB.

Genetic variation between samples

Exact tests for pairwise genic differentiation [50] were performed with GENEPOP ON THE WEB [47,48]. The significance levels of those tests were corrected with Benjamini and Hochberg's [51] false discovery rate procedure when necessary as those tests can involve non orthogonal and multiple comparisons. Genetic differentiation between pairs of samples was summarised by Weir and Cockerham's [52] estimator of pairwise $F_{\rm ST}$ using GENEPOP ON THE WEB.

Bottleneck tests

For each field population and laboratory colony, tests for a recent reduction in population size in the last 2Ne-4Ne generations were performed using the program BOTTLENECK 1.2

[53,54]. These analyses were carried out assuming a generalised stepwise mutation (GSM) with a variance of 0.36 [31,55,56]. One-tailed Wilcoxon sign-rank tests were used to determine whether observed heterozygosity deviates from expectations at mutation-drift equilibrium. Estimations were based on 10000 replications. Reductions in population size were also tested using the "mode-shift" indicator of the distortion of allele frequency classes' distributions [57].

Assignment of the IAEA individuals to the potential source populations

Two approaches were used. In the first, we calculated the mean multilocus individual assignment likelihood of each IAEA sample *i*, to each sample of possible source populations *s* [hereafter denoted $L_{i\rightarrow s}$, see 58,59] with GENECLASS 2 ver. 2.0.g [46]. For each IAEA sample, the most probable source population was then identified as that with both the highest $L_{i\rightarrow s}$ value and the lowest $F_{\rm ST}$ value with the source population considered [24,60].

The second method allowed the concomitant assignment of individuals and inference of potential admixture. This clustering approach, implemented in STRUCTURE 2.3.4 [61] was used to evaluate the contribution of the Rukomeshi and Busia populations to the current IAEA colony. Individual multilocus genotypes were used to infer clusters of individuals within which deviation from HWE and linkage disequilibria are minimized. The microsatellite data were converted from GENEPOP to STRUCTURE format using the software CREATE v.1.37 [62]. Ten replicate runs for each prior value of the number (K) of clusters, set between 1 and 5, with a burn-in of 2×10^5 iterations followed by 10^6 iterations. The admixture model of ancestry together with the correlated allele frequencies model were used [63] and no account was taken a priori on the origin (Busia, Rukomeshi or IAEA) of each individuals, i.e. individuals were clustered only on the basis of their multilocus genotypes. Default values were maintained for all other parameters. K was estimated as the value leading to the highest likelihood for the data P(X | K) and with the ΔK statistics of Evanno et al. [64] with STRUCTURE HARVESTER Web v0.6.93 [65].

Inferring the past demography of the laboratory colony using Approximate Bayesian Computation

We applied an Approximate Bayesian Computation (ABC) approach to infer the demographic history of the *G. pallidipes* IAEA colony and field populations under study. Microsatellite data were combined with prior information on the history and demography of those populations. Analyses were performed with DIVABC v 1.0.4.46 [66,67]. Briefly, in an ABC analysis, summary statistics of each simulated dataset are recorded, together with the label of the scenario used for the simulation. Euclidian distances between each simulated dataset and the observed dataset are computed. These distances are then used to estimate the posterior probabilities of

the scenarios and posterior probability distributions of the parameters. In each of the three analyses described below and in Supplementary file S2, 10^6 datasets were simulated for each competing scenario using parameter values drawn from prior distributions and assuming equiprobability of each scenario *a priori*. The simulated datasets had the same characteristics (number of samples, individuals, loci, characteristics of the microsatellite loci) as the observed dataset.

Genetic variation was summarised using a set of summary statistics traditionally used in ABC for each population and each population pair [23,34,67]: mean number of alleles, mean gene diversity, mean allele size variance and mean M index across loci [68], pairwise $F_{\rm ST}$ [52], mean individual assignment log-likelihoods of individuals from population *i* assigned to population *j* ($L_{i\rightarrow j}$) and the maximum likelihood estimates for admixture proportions [69]. In analyses 1 and 2, four summary statistics were used while there were 54 in analysis 3 (Table S3).

In analysis 1 (Supplementary file S2) we focused on the Busia population in order to correctly model the demographic history of this population when analysing the IAEA colony history. This is of importance as the Busia population may have experienced a genetic bottleneck due to tsetse control [70] or to the destruction of the tsetse habitat associated with the increase of the human population between the foundation of the IAEA colony (1975) and the sampling of the Busia population. If such a bottleneck occurred it is important to take it into account when performing inferences on the demographic history of the IAEA population.

In analysis 2 (Supplementary file S2) we focused on the demography of the Rukomeshi population between the establishment of the IAEA colony and the sampling of the Rukomeshi files in 2006. Unlike for the Busia population, there is no record of any tsetse control program in Rukomeshi area between 1975 and 2006. However, a field trial of a tsetse control technique has been carried out in Rukomeshi in 1991 and could have decreased the size of the *G. pallidipes* population temporarily [71].

The IAEA colony demography and origin were examined in analysis 3, taking into account the scenarios selected in analysis 1 and 2. The IAEA colony was considered to originate from a single source, Busia or Rukomeshi, or from an admixture between both. Each of those three scenarios were considered with or without the possibility of a bottleneck associated with the laboratory establishment of the IAEA colony, giving a total of 6 competing scenarios (Figure 1). The analyses were performed using parameter values drawn from the prior distributions described in Table 2.

For all the ABC analyses performed, posterior probabilities of the competing scenarios were estimated by polychotomous logistic regression [67] on the 1% simulated datasets closest to the observed dataset. The selected scenario was that obtaining the highest posterior probability with a 95% confidence interval nonoverlapping with the second highest probability [30,31]. The posterior distributions of the demographic parameters were estimated under the selected scenario using a local linear regression on the 1% simulated datasets producing the smallest Euclidian distances to the observed dataset [27,67]. The median of a posterior distribution was considered as point estimate for a parameter [as in 66,67].

ABC analyses were performed on simulated pseudo-observed datasets (PODs) to evaluate the ability of our ABC analysis 3 to select the true scenario. For each of the 6 scenarios of the ABC analysis 3 (Figure 1), 100 PODs were simulated using parameter values drawn from the probability distributions identical to the prior distributions (Table 2). Each PODs has the same characteristics (number of samples, individuals, loci) as the observed dataset. For the selection of the scenario, procedures previously described

(summary statistics, Euclidian distances, posterior probability estimation) were applied to each POD. Because the scenario used to generate each POD is known, applying the ABC analysis 3 on the PODs allows the estimation of type I and II errors for these analyses. Type I error corresponds to the proportion of PODs for which a scenario is excluded by the ABC analysis while it is actually the true scenario (the one used to generate the PODs). Type II error corresponds to the proportion of PODs for which a scenario is selected while it is not the true one. Low type II error indicates that the results are reliable even when the type I error is large [23].

Using the "model checking" option in DIYABC we evaluated the ability of the selected scenario and of its parameters posterior distributions to generate simulated data that are similar to the observed data set [i.e. evaluation of the "goodness of fit" of the combination of the selected scenario and its parameter posterior distributions, 66]. The procedure was carried out by simulating 10⁴ PODs using the scenario selected in the ABC analysis 3 and parameters values drawn from the posterior distributions of the parameters. Summary statistics distributions corresponding to those 10⁴ PODs were then compared to the observed summary statistics. To reduce the bias introduced by the use of the same set of summary statistics for the ABC analysis and the model checking [66] we added the following summary statistics to the previously used 54 summary statistics: the shared allele distance [72] and the $(\delta \mu)^2$ distance $[(\delta \mu)^2 = (\mu_A - \mu_B)^2$, where μ_A and μ_B are the "means of allele size in populations A and B", 73]. That way, 66 summary statistics were used in the "model checking". The combination of the selected scenario and its parameter posterior distributions would be considered inadequate if many observed summary statistics were not included in the distribution of the summary statistics corresponding to the 10^4 PODs [66].

Simulation of the genetic diversity of the source population of the IAEA colony in 1975

In order to evaluate the genetic diversity loss that occurred between the 1975 source population and the current IAEA colony we used DIYABC to simulate the source population in 1975 under the scenario selected by the ABC analysis 3. To take into account the inter-simulation variation, we simulated 100 datasets under the scenario selected by the ABC analysis 3. In addition to the four samples (Busia, Rukomeshi and IAEA 2012 and 2013) included in the ABC analysis 3, a fifth sample corresponding to the 1975 source population was simulated. This sample of the 1975 source population correspond to 26 multilocus genotypes at 9 microsatellite loci having the same characteristics as the ones used to produced the observed data. We simulated 26 multilocus genotypes to be able to directly compare the genetic diversity of the 1975 source population simulated and of the IAEA 2013 sample which is made up of 26 individuals (Table 1). The simulations were performed using the parameter values estimated previously under the scenario selected by the ABC analysis 3 (i.e. the median of the posterior distributions of the parameters). Because the simulations slightly over estimated the genetic diversity of the Rukomeshi sample we downscaled the genetic diversity simulated for Busia 1975 sample accordingly to be conservative in our comparison of IAEA 2013 and the simulated Busia 1975 sample.

Results

The microsatellite locus GmmF10 could be amplified in each of the flies samples in Busia (see Figure S1) indicating that they are all *G. pallidipes* [38,39].



Figure 1. Competing scenarios considered in the ABC analysis of IAEA colony past demography (analysis 3). The demography of the Busia and Rukomeshi populations was determined as described in Supplementary file S2. In all scenarios, the two potential source populations merged $T_{anc.}$ ago into an unsampled ancestral population and a bottleneck started in Busia in 1991 and lasted $BD_{cont.}$ generations. Scenario 4, 5 and 6 are respectively variations of scenarios 1, 2 and 3 in which a genetic bottleneck (of duration $BD_{col.Bus.}$ or $BD_{col.Ruk.}$) is associated with the laboratory colonisation. In scenarios 1, 2, 4 and 5 the IAEA colony has a single population of origin. In scenarios 1 and 4, the IAEA colony was founded from the Busia population in 1975 while in scenarios 2 and 5, it was founded from the Rukomeshi population $T_{col.Ruk.}$ generations ago. In scenario 3 and 6 the IAEA colony originates from an admixture between unsampled colonies of Busia and Rukomeshi origin that were respectively founded from the Busia population in 1975 and from the Rukomeshi population $T_{col.Ruk.}$ generations ago. When admixture occurs, the admixture rate *ar* is the proportion from unsampled Busia colony that contributed to the admixed IAEA colony. doi:10.1371/journal.pntd.0002697.g001

Genetic variation within samples

The complete dataset analysed (121 individuals from Busia, Rukomeshi and the IAEA colony) showed moderate polymorphism with an average of 6.556 (SD = 3.046) alleles per locus over all samples. The number of alleles ranged from 3 at locus GmmC17 to 11 at loci GpC101 and GmmA06. Fifty seven (96.6%) of the 59 alleles observed over all samples were present in the Rukomeshi sample. In Rukomeshi all loci were polymorphic whereas locus GpCAG133 was monomorphic in Busia and both IAEA samples and locus GmmC17 was monomorphic in IAEA 2013. Only the Rukomeshi sample was found to deviate from Hardy-Weinberg equilibrium (Table 1) which was mainly due to a single locus. Only allele frequencies at locus Gpc26b were found to significantly deviate from HWE in Rukomeshi after correction for multiple comparisons (p<0.001). This was associated with heterozygote deficiency (positive $F_{\rm IS}$) and could be the results of the presence of null alleles at this locus in Rukomeshi. Most of the analyses including the STRUCTURE analysis and the ABC analysis 3 were repeated after excluding locus Gpc26b from the dataset. The results obtained were very similar to the ones obtained with 9 loci (data not shown) and the deviation from HWE at locus Gpc26b was thus considered inconsequential for our inferences. No linkage disequilibrium was detected between loci in any sample.

The mean number of alleles per locus (*Na*) was heterogeneous between samples, ranging from 2.2 [Allelic richness (AR) = 2.1] in the IAEA colony to 6.3 (AR = 5.9) in Rukomeshi (Figure 2). Similarly, mean expected heterozygosity (*H*) ranged from 0.19 in the IAEA colony to 0.56 in Rukomeshi (Figure 2). *Na* and *AR* were significantly higher in Rukomeshi than the other three samples (p < 0.01, 9 signed ranks, |W| = 45) while they were no different in Busia and the IAEA samples $(p > 0.05, \le 8 \text{ signed ranks}, |W| \le 18)$. A different pattern was observed with H which was significantly higher in the field populations than in the IAEA colony $(p \le 0.01, \ge 8 \text{ signed ranks}, |W| \ge 34)$ but was not different between the two field populations $(p \ge 0.05, 9 \text{ signed ranks}, |W| = 29)$. The tests for mode shift of the distribution of allele frequency classes and for heterozygosity excess (Figure 3) indicated that the Busia population recently experienced a genetic bottleneck while those tests revealed no evidence of such bottlenecks in Rukomeshi or IAEA populations.

Genetic variation between samples

Pairwise comparisons between samples showed highly significant genetic differentiation tests and large to very large $F_{\rm ST}$ estimates (mean = 0.27, SD = 0.13; Table 3) except for the comparisons between the IAEA colony samples (no genetic differentiation and $F_{\rm ST}$ = 0; Table 3). The genetic differentiation between Rukomeshi and the IAEA samples was significantly greater than between the Busia and the IAEA samples (Wilcoxon' sign rank test on pairwise $F_{\rm ST}$ per locus, 8 sign ranks, |W| > 28, p < 0.05).

We also investigated the mean multilocus individual assignment likelihood of each IAEA sample *i* to each sample of the potential source population *s* ($L_{i\rightarrow s}$) and the results are indicated in Table 3. The highest $L_{i\rightarrow s}$ values for Busia also indicate that this is a more likely source population of the IAEA colony than Rukomeshi.

Table 2. Prior distributions of the parameters used in the ABC analyses.

Parameters	Prior distributions
N _{anc.} , N _{Bus.} , N _{Ruk.}	Uniform [500; 20000]
N _{IAEA} , N _{col.Bus.} , N _{col.Ruk.}	Uniform [1; 20000]
NB _{col.Bus} , NB _{col.Ruk} , NB _{cont.}	Uniform [1; 100]
T _{anc.} (generations before present)	Uniform [500; 25000]
T _{col.Ruk.}	Uniform [1980; 1983]
T _{admix.}	Uniform [1981; 1990]
BD _{cont.}	Uniform [1,47]
BD _{col.Bus.}	Uniform [1,15]
BD _{col.Ruk.}	Uniform [1,25]
ar	Uniform [0.1; 0.9]
Mutational model	GSM+SNI with default parameters
Number of generations per year	5*

Note: $N_{anc.}$: effective population size (*Ne*) of the ancestral population. $N_{Bus.}$: *Ne* of the Busia population. $N_{Ruk.}$: *Ne* of the Rukomeshi population. N_{IAEA} : *Ne* of the IAEA colony. $N_{col,Bus.}$: *Ne* of the unsampled colony of Busia origin. $N_{col,Ruk.}$: *Ne* of the unsampled colony of Rukomeshi origin. $N_{col,Bus.}$: *Ne* of the colony of Busia origin (JAEA in scenario 4 and unsampled in scenario 6) during the bottleneck associated with its establishment that started in 1975 lasted $BD_{col,Bus.}$

generations. $NB_{col.Ruk}$: Ne of the colony of Rukomeshi origin (IAEA in scenario 5 and unsampled in scenario 6) during the bottleneck associated with its establishment that started at $T_{col.Ruk}$, and lasted $BD_{col.Ruk}$. generations. NB_{cont} : Ne of the Busia population during the bottleneck associated with the tsetse control that started in 1991 and lasted BD_{cont} . generations. T_{anc} : number of generations between present and the merge of the two potential source populations an unsampled ancestral population. T_{admix} : date of the admixture. ar: admixture rate. GSM: Generalized Stepwise Mutation model. SNI: Single nucleotide indel

mutations. *: see Supplementary file S2 (Table S7 in Supplementary file S2) for a rationale. doi:10.1371/journal.pntd.0002697.t002

In a third approach, STRUCTURE inferred the most likely number of genetic clusters (*K*) to be 2 or 3. The statistic ΔK [64] indicated K=2 while the highest likelihood of the data was obtained at K=3(Figure 4A). At K=2 (Figure 4B, top) all the IAEA individuals belong to the same cluster as the individuals from Busia. Also a very small fraction of the multilocus genotypes of some IAEA individuals clustered with Rukomeshi but this fraction was smaller for the IAEA colony than for Busia and was thus considered uninformative. Similarly, there was no evidence of introgression from Rukomeshi to the IAEA colony at K=3 (Figure 4B, bottom). At K=3 each population (Busia, IAEA colony and Rukomeshi) formed a distinct cluster except for a few Busia individuals that clustered within the IAEA cluster or were admixed between the Busia and the IAEA cluster and vice and versa.

Manual inspection of the allele frequency distributions did however support the possibility of introgression from Rukomeshi into the IAEA colony. All the alleles present in the IAEA samples should normally be present in the sample of its source population and all the alleles of the IAEA colony were found in Rukomeshi but not in Busia. 4 alleles observed in the IAEA colony were absent from the Busia sample and present in Rukomeshi. Two of these alleles (allele 129 at locus GmmK06 and allele 187 at locus GmmC17) were rare in the IAEA colony (allele frequencies ≤ 0.02) while the other two (alleles 157 and 159 at locus GpA19a) had relatively high frequencies (0.04 to 0.29). However, the interpretation of these data must also take into account the total number of alleles in both Busia and Rukomeshi (23 and 57 respectively). Fisher's exact test shows that the number of private alleles from Busia observed into the IAEA colony (0/23) is not statistically different from the number of private alleles from Rukomeshi (4/57; p = 0.32).

Inferring the past demography of the laboratory colony using Approximate Bayesian Computation

In the ABC analysis 1 focused on Busia, the scenario with a strong bottleneck (scenario A) obtained a posterior probability of 0.997 (Table 4). The confidence in this scenario choice is good because type II error was small (0.065, Table 4). In the ABC analysis 2 focused on Rukomeshi, the highest posterior probability (0.737) was associated with the scenario with a constant population size (scenario C). However the power of the ABC analysis 2 was low with high type I (0.66) and type II errors (0.275). It should be noted that 84.8% of type I error correspond to the selection of scenario B when scenario C is true and 87.3% of type II error is associated with the selection of scenario B is true. To conclude, the results of the ABC analyses 1 and 2 show that a genetic bottleneck occurred recently in Busia but provide no support for such a bottleneck in Rukomeshi.



Figure 2. Genetic variations within samples. Error bars indicate the standard deviations across loci. *Na*: average number of alleles per locus. *AR*: allelic richness. *AR* is based on minimum sample size (N = 23 in Friuli for locus DVV-ET1). *H*: mean expected heterozygosity. doi:10.1371/journal.pntd.0002697.g002



Figure 3. Distribution of the allele frequency classes. (A) in Busia. (B) in the IAEA colony samples from 2012 and 2013. (C) in Rukomeshi. The result of the bottleneck tests (Wilcoxon' tests on heterozygosity excess, and the mode shift tests) are indicated for each panel which correspond to a field population or to the IAEA colony. Heterozygosity excess and/or a shifted mode in the distribution of allele frequency classes indicate a recent reduction in population size. doi:10.1371/journal.pntd.0002697.g003

Those results were taken into account and the demography of Busia and Rukomeshi were modelled accordingly in the ABC analysis 3 (Figure 1). To identify the most likely source population of the IAEA colony and estimate the intensity of genetic drift associated with its foundation and establishment, six possible scenarios were considered (Figure 1). Scenario 4 involving a single Busia origin of the IAEA colony and a genetic bottleneck **Table 3.** Genetic differentiation between samples and assignment of the IAEA colony samples into the potential source populations.

	Busia	IAEA-2012	IAEA-2013	Rukomeshi
Busia	-	4.0×10 ⁻⁵	2.8×10 ⁻⁵	-
IAEA-2012	0.133	-	-	1.9v10 ⁻¹²
IAEA-2013	0.130	0	-	4.3×10 ⁻¹²
Rukomeshi	0.276	0.406	0.389	-

Pairwise F_{ST} are indicated below the diagonal. Significant pairwise genotypic differentiation exact tests are indicated in bold typeface. The mean individual assignment likelihoods of IAEA samples into the potential sources of the colony ($l_{i\rightarrow s}$) are indicated above the diagonal. doi:10.1371/journal.pntd.0002697.t003

associated with its foundation and establishment was identified as having the highest posterior probability (0.792, Table 4) with small type II error (0.044, Table 4). Although the type I error associated with this analysis was not small (0.25, Table 4), it should be noted that 76% of this corresponds to the selection of scenario 1 when scenario 4 is true (i.e. selection of a scenario in which the source population of the IAEA colony is Busia). The model checking analysis revealed that the combination between scenario 4 and the estimated posterior distributions of the parameters produced simulated data very similar to the observed data. Indeed, only one observed summary statistic [the $(\delta \mu)^2$ distance, 73] between the IAEA samples (which was not used in our ABC analyses) was in the 5% tail of the distribution of the corresponding simulated statistics. These results give good confidence in the estimation of the parameters associated with scenario 4.

Posterior probability distributions of the effective population sizes (Ne, Figure 5) illustrate the intensity of genetic drift in the field populations and of the IAEA colony. N_{Bus} . corresponds to Busia Ne before the bottleneck; the genetic data for the Busia sample did not contain any information about the Busia Ne after the genetic bottleneck (data not shown). All the posterior distributions of the $\mathcal{N}e$ estimated substantially differ from the corresponding prior probability distributions (Figure 5) which indicates that the empirical data collected in Busia, Rukomeshi and the IAEA colony contain information about the estimated Ne. The estimated Ne for Busia was the smallest and similar to that for the IAEA colony. In contrast, the Ne estimated for Rukomeshi was an order of magnitude higher. A small effective population size was associated with the foundation and establishment of the IAEA colony and this genetic bottleneck was associated with an important loss of genetic diversity in the IAEA colony compared to its source population in 1975 (35.66% and 53.99% loss of Na and H respectively, p<0.05, 9 signed ranks, $|W| \ge 35$, Figure 6).

Discussion

The origin of the IAEA colony

The *G. pallidipes* IAEA colony is the only one available for mass rearing of this important vector of human and animal African trypanosomiasis. It is the source population for Sterile Insect Technique (SIT) programs in Kenya, Tanzania and Ethiopia [13]. In the present study we have analysed the genetic variation within and between the IAEA colony and its two potential source populations and combined it with historical information to identify the most likely source population and infer the past demography of the colony.



Figure 4. Estimated number of clusters and population structure from the STRUCTURE analysis. (A) Mean (\pm SD) natural logarithm of the likelihood of the data [LnP(X|K)] over 10 Structure replicated runs for each value of the putative number of clusters (K). (B) Estimated population structure from the STRUCTURE analysis for K = 2 and 3. Each individual is represented by a vertical line divided into K coloured segments that represent the individual's estimated membership fractions in K clusters. Black lines separate individuals from different samples. Each plot, is based on the highest-probability run (among ten) at K = 2 and 3. doi:10.1371/journal.pntd.0002697.g004

Measurements of genetic variation between samples and Bayesian clustering clearly indicated a strong similarity between the colony and the Busia population and no introgression from Rukomeshi. The only counter-evidence is four private alleles from Rukomeshi (present in Rukomeshi and not in Busia) that were observed in the IAEA colony. This result lacked statistical significance however. A potential explanation for this observation is that the Busia population experienced a genetic bottleneck between the establishment of the IAEA colony and the collection of our Busia sample. Those four alleles could have been lost in the Busia population through this genetic bottleneck which would imply that, while those four alleles are private to Rukomeshi relatively to our Busia sample, they might not have been private to Rukomeshi relatively to the Busia population before the genetic bottleneck.

The bottleneck in Busia is most likely to have been caused by tsetse control measures carried out in the area in the 1990's [70] and thus limit the utility of our Busia sample for drawing robust conclusions about the intensity of the genetic drift associated with the foundation and establishment of the IAEA colony in the 1970's through direct comparison of the genetic variation within samples. To overcome this restriction an ABC approach allowed inferences about the past demography of the IAEA colony based on contemporary samples of the colony and of its potential source population. The results suggest that taking into account the past demography of the source population was very relevant in the context of this study as it allowed us to make more accurate inferences about the intensity of the genetic drift that was associated with the foundation and establishment of the IAEA colony. Such inferences would have been biased in a direct comparison of the Busia sample and of the IAEA samples because

the Busia population experienced a genetic bottleneck between the foundation of the IAEA colony and the sampling in Busia.

The ABC analysis gives a numerical prediction for the population size during the bottleneck. Our estimates revealed that the genetic drift associated with the foundation and establishment of the IAEA colony was strong with an effective number of individuals associated with the foundation and establishment of the colony around 27. This is consistent with the historical record that only 36 females were used to start the colony [14].

To conclude, all the analyses indicate clearly that the area of origin of the IAEA colony is the Kenya/Uganda border and that a severe genetic bottleneck was associated with the foundation and establishment of the colony causing a marked loss of genetic diversity. These data contrast with earlier findings on colonies of *G. pallidipes* but those studies were constrained by using isoenzyme markers and analytical approaches based largely on measurement of allele frequencies and heterozygosity [5,74].

Illustration of the usefulness of model based inferences in population genetics

The data set analysed in the present study is a good illustration of the usefulness of model based inferences, such as the ABC, over more classical population genetics analyses. It is clear from our data that a direct comparison of the genetic diversity of laboratory colony with that of the sample collected from its source population resulted in similar values for *Na* and AR which, in isolation, seem to indicate no loss of diversity in the colony. These findings contrast however with the heterozygosity excess, the mode shift of the distribution of the allele frequency classes and the ABC analysis all of which indicate that the Busia population has

Table 4. Posterior probability of the selected scenarios and confidence in scenario choice in the ABC analyses.

ABC analysis	Number of competing scenarios	Selected scenario	Posterior probability [95% CI]	Type I error	Type II error Mean (min-max)
1 - Busia	3	Scenario 1 (strong bottleneck)	0.997 [0.997,0.998]	0.17	0.065 (0.05–0.08)
2 - Rukomeshi	3	Scenario 3 (constant population size)	0.737 [0.724,0.750]	0.66	0.275 (0.07–0.48)
3 – IAEA colony	б	Scenario 4 (Busia origin with bottleneck)	0.792 [0.756,0.827]	0.25	0.044 (0.05–0.17)

doi:10.1371/journal.pntd.0002697.t004



Figure 5. Effective population sizes (*Ne*) **estimated from the ABC analysis with scenario 4 (Figure 1).** The prior distributions are shown as black lines while the posterior distributions are shown as coloured lines using the same colour code as in Figure 1. The medians of a posterior distribution, considered as point estimate of the parameters, are indicated. Each distribution was obtained from 10000 values. (A) $N_{Bus.}$: *Ne* of the Busia population. $N_{Ruk.}$: *Ne* of the Rukomeshi population. $N_{IAEA.}$: *Ne* of the IAEA colony. The prior distributions for $N_{Bus.}$ and $N_{Ruk.}$ are shown as a plain line while the prior distribution for $N_{IAEA.}$ is shown as a dashed line. (B) $NB_{col.Bus.}$: *Ne* of the colony of Busia origin (which is at the origin of the current IAEA colony) during the bottleneck associated with its establishment. doi:10.1371/journal.pntd.0002697.q005

experienced a severe genetic bottleneck. Using the ABC it was possible to take into account the demography of the Busia population between the foundation of the IAEA colony and the sampling of flies in Busia. The current study clearly demonstrates that model based inferences, such as ABC, are more powerful at detecting genetic bottlenecks compared to moment based methods such as the ones implemented in the program BOTTLENECK [54] that was unable to detect any bottleneck in the IAEA colony from contemporary samples. This observation accords with previous findings both empirical [e.g. 75] and by simulation [76].

As indicated by Luikart et al. [57] and Piry et al. [54] the tests implemented in the program BOTTLENECK are able to detect recent genetic bottleneck 2Ne-4Ne generations before sampling. The bottleneck associated to the IAEA colony is likely to have occurred about 150 generations before sampling which potentially explains why the tests implemented in BOTTLENECK did not detect any



Figure 6. Loss of genetic diversity in the IAEA colony (2013 sample) with respect to its simulated source population at the Kenya/Uganda border in 1975. Na; allelic diversity loss and H; expected heterozygosity loss. Both losses are statistically significant (Wilcoson's signed-rank tests over loci, $p \le 0.05$). doi:10.1371/journal.pntd.0002697.g006

bottleneck in the IAEA colony from contemporary samples. Moreover, Hoban et al. [77] recently showed by simulation that a "recovery after a moderate amount of time" is associated with an important reduction of power in bottleneck detection when using moment based methods. This finding is consistent with a rapid recovery after bottleneck that occurred in the IAEA colony from 36 to over 1500 females in approximately three years [14].

Potential consequences for Sterile Insect Technique (SIT) programs

The high level of genetic drift that occurred in the IAEA colony could have a negative impact on SIT because some of the wild-type characteristics of *G. pallidipes* could potentially have been lost during the strong genetic bottleneck that was associated with the IAEA colony foundation and establishment. For example, some genetic diversity associated with mating competitiveness could have been lost that would lead to a suboptimal mating competitiveness of the colony males in the field.

Theoretically, several things can be done to reduce genetic drift associated with rearing organisms in captivity. In a colony already established it is possible to limit genetic drift by reducing the variation in reproductive success by reducing the number of nonmated individuals and/or by homogenising the contribution of each family to the next generation. However, for SIT, that requires mass production of males for sterilization and release into to field, reducing the variation in reproductive success is not feasible. Indeed, SIT requires the production of tens of thousands of insects and, in such conditions, the work load associated with measures seeking the homogenisation of the contribution of each family to the next generation would be too demanding. Moreover, for tsetse SIT, the requirement to produce spare males to be sterilised goes against any procedure that would aim at reducing the number of non-mated individuals. Indeed, males for SIT are spared before mating (and some techniques even seek to identify the males during the pupal period) and this is made possible by the use of a skewed sex ratio at mating [1 male to 3 or 4 females depending on tsetse species, 78].

For colony established from scratch, it is obvious that maximising the number of founder individuals and/or providing a continuous input of individuals from the field will limit the genetic diversity loss compared to the field population of origin. However, this could also prevent laboratory establishment and mass rearing that requires a high level of adaptation of the reared organisms to the laboratory conditions [4]. Indeed a continuous input of flies from the laboratory to the field would lead to a continuous gene flow from the wild to the laboratory and could prevent adaptation to the laboratory (i.e. the increase in frequency of "laboratory adapted" gene combinations).

In G. pallidipes, many unsuccessful attempts have been made to raise small laboratory colonies [e.g. 79] as well as larger mass rearing colonies [80-82]. Although some of those unsuccessful attempts could be due to fly diseases [82], genetic effects could also provide explanations. Surprisingly, compared to other G. pallidipes colonies reported in the literature, the IAEA colony was started from the smallest number of founding individuals while it is the only one that has reached mass rearing. Indeed, all the attempts to establish G. pallidipes colonies that are reported in the literature have been made with many more founding individuals [see 79, for failure, 79 & 83, for maintenance of very small colonies and 84 & 85 for successful colonisations]. The 36 producing females that were at the origin of the IAEA colony [14] may have carried a gene pool particularly suitable for laboratory colonisation. Alternatively it is possible that some deleterious alleles have been purged by genetic drift [86,87], making this colony more successful than others.

Combining our results with those of Ouma et al. [35] allows evaluation of the representativeness of the IAEA colony compared to G. pallidipes field populations. When doing so it is important to take into account the fact that the Busia population likely experienced a genetic bottleneck, probably in the 1990's. In accordance with our inferences about this genetic bottleneck in Busia, Ouma et al. [35] have shown that the genetic diversity of the Busia population is on average around 25% (32% for AR and 21% for H lower than that of other G. pallidipes populations in Uganda and Kenya. Taking into account this discrepancy of genetic diversity between Busia and other field populations, we can conclude that the genetic diversity of the IAEA colony is on average around 55% (45% for AR and 63% for H) lower than that of other G. pallidipes populations in Uganda and Kenya. This is concordant with our comparison between the genetic diversity of the IAEA colony with the simulated Busia 1975 sample (Figure 6). The differences between the genetic diversity present in the IAEA colony and populations that are genetically highly diverse like Rukomeshi [22] is of course even larger. Indeed, the genetic diversity in the IAEA colony is around 65% (65% for AR and 66% for H lower than in the Rukomeshi population.

Insect laboratory colonies should be genetically similar to field populations for meaningful studies to be performed on different aspects of biology of a species [e.g. 5]. In this respect, our data imply that results obtained using the IAEA colony should be interpreted with some caution with regard to the biology of *G. pallidipes*. It should be noted that no genetic diversity loss compared to field populations have been detected in other colonies of

References

- Holmes P (2013) Tsetse-transmitted trypanosomes Their biology, disease impact and control. J Invertebr Pathol 112: S11–S14.
- Vreysen MJB, Salch KM, Lancelot R, Bouyer J (2011) Factory Tsetse Flies Must Behave Like Wild Flies: A Prerequisite for the Sterile Insect Technique. PLoS Neglect Trop D 5: e907.
- Klassen W, Curtis CF (2005) History of the Sterile Insect Technique. In: Dyck VA, Hendrichs J, Robinson AS, editors. Sterile Insect Technique: Springer Netherlands. pp. 3–36.
- Krafsur ES (2005) Role of Population Genetics in the Sterile Insect Technique. In: Dyck VA, Hendrichs J, Robinson AS, editors. Sterile Insect Technique: Springer Netherlands. pp. 389–406.

different species of tsetse (G. tachinoides, G. m. morsitans, G. m. centralis and G. pallidipes) except in a G. p. gambiensis colony [reviewed in 5].

Supporting Information

Figure S1 Successful amplification of locus GmmF10 in all the samples from Busia. (PDF)

Supplementary file S1 Evaluation of 15 microsatellite loci for their use in population genetics analyses of *G. pallidipes* using multiplex PCR. Includes **Table S4** (*G. pallidipes* samples used in the evaluation of the 15 microsatellite loci), **Table S5** (Characteristics of the 15 microsatellites loci evaluated) and **Table S6** (Validation of the microsatellite loci for their use for population genetics). (DOCX)

Supplementary file S2 Inferring the past demography of the Busia and the Rukomeshi populations using Approximate Bayesian computation (ABC). Includes **Table S7** (Prior distributions of the parameters used in the Busia ABC analysis) and **Figure S2** (Demographic scenario considered to infer the demographic history of the Busia and Rukomeshi populations). (DOCX)

Table S1Characteristics of the 9 microsatellites loci used in thisstudy.

(DOCX)

Table S2Allele "bins" defined for the 9 G. pallidipes microsatellite loci used.

(DOCX)

 Table S3
 Summary statistics used in the different ABC analyses.

 (DOCX)
 Image: State of the sta

Acknowledgments

We thank Anne-Marie Donachie and Alana Hamilton (University of Glasgow, UK) as well as Esther Waweru and James Kabii (ICIPE, Kenya) for technical and administrative assistance; Michael Miles and James Patterson (London School of Hygiene and Tropical Medicine, UK), Johnson Ouma (KARI-Trypanosomiasis Research Centre, Kenya) and Adly Abd-Alla (Insect Pest Control Laboratory, FAO/IAEA, Austria) for their assistance with acquisition of field samples; Andrew Parker (Insect Pest Control Laboratory, FAO/IAEA, Austria) for supplying the IAEA colony samples and for his research regarding the history of the colony; Eric Lombaert (INRA, France) for his advice regarding the ABC analyses and Anne Fischer (ICIPE, Kenya) for critical reading of an earlier version of this manuscript.

Author Contributions

Conceived and designed the experiments: MC DKM CMRT. Performed the experiments: MC. Analyzed the data: MC. Contributed reagents/ materials/analysis tools: MC DKM CMRT. Wrote the paper: MC DKM CMRT.

- Gooding RH (1990) Genetic aspects of quality control in tsetse colonies. Insect Sci Appl 11: 385–398.
- Vreysen MJB, Saleh KM, Ali MY, Abdulla AM, Zhu ZR, et al. (2000) Glossina austeni (Diptera : Glossinidae) eradicated on the Island of Unguja, Zanzibar, using the sterile insect technique. J Econ Entomol 93: 123–135.
- Gibson WC, Wellde BT (1985) Characterization of *Trypanozoon* stocks from the South Nyanza sleeping sickness focus in Western Kenya. Trans Roy Soc Trop Med Hyg 79: 671–676.
- Auty HK, Picozzi K, Malele I, Torr SJ, Cleaveland S, et al. (2012) Using Molecular Data for Epidemiological Inference: Assessing the Prevalence of *Trypanosoma brucei rhodesiense* in Tsetse in Serengeti, Tanzania. Plos Neglect Trop Dis 6: e1828.

- Njiru ZK, Ndung'u K, Matete G, Ndungu JM, Gibson WC (2004) Detection of Trypanosoma brucei rhodesiense in animals from sleeping sickness foci in East Africa using the serum resistance associated (SRA) gene. Acta Trop 90: 249–254.
- Njiru ZK, Makumi JN, Okoth S, Ndungu JM, Gibson WC (2004) Identification of trypanosomes in *Glossina pallidipes* and *G. longipennis* in Kenya. Infect Genet Evol 4: 29–35.
- International Atomic Energy Agency IAEA (2011) Strengthening the Agency's Activities related to Nuclear Science, Technology and Applications.
- Parker A (2013) Joint FAO/IAEA Insect Pest Control Laboratory, Vienna, Austria, pers. com.
- Anonymous (2013) World-Wide Directory of SIT Facilities (DIR-SIT), joint FAO/IAEA Programme of Nuclear Techniques in Food and Agriculture, Available at: http://nucleus.iaea.org/dirsit/dirsitx.aspx [Accessed January 2014].
- 14. Leegwater van der Linden ME (1980) Recent advances in the rearing of *Glossina pallidipes* Austen [413–422]. In: Proceedings of the International Symposium on the Use of Isotopes for Research and Control of Vectors of Animal Diseases, Host Pathogen Relationships and the Environmental Impact of Control Procedures; 7–11 May 1979; Vienna, Austria.
- Abd-Alla A, Bossin H, Cousserans F, Parker A, Bergoin M, et al. (2007) Development of a non-destructive PCR method for detection of the salivary gland hypertrophy virus (SGHV) in tsetse flies. J Virol Methods 139: 143–149.
- Briceno RD, Eberhard WG, Robinson AS (2007) Copulation behaviour of *Glossina pallidipes* (Diptera : Muscidae) outside and inside the female, with a discussion of genitalic evolution. B Entomol Res 97: 471–488.
- Mutika GN, Opiyo E, Robinson AS (2002) Effect of low temperature treatment on the quality of male adult *Glassina pallidipes* (Diptera: Glossinidae) in relation to the sterile insect technique. Entomol Sci 5: 209–214.
- Mutika GN, Parker AG (2006) Induced sterility of *Glossina pallidipes* Austen males after irradiation in a nitrogen atmosphere. Entomol Sci 9: 47–53.
- Olet PA, Opiyo E, Robinson AS (2002) Sexual receptivity and age in *Glossina* pallidipes Austen (Dipt., Glossinidae). J Appl Entomol 126: 86–91.
- Dyer NA, Lawton SP, Ravel S, Choi KS, Lehane MJ, et al. (2008) Molecular phylogenetics of tsetse flies (Diptera: Glossinidae) based on mitochondrial (COI, 16S, ND2) and nuclear ribosomal DNA sequences, with an emphasis on the palpalis group. Mol Phylogenet Evol 49: 227–239.
- 21. Langley P (2013) Tsetse Research Laboratory, Bristol, UK, pers. com.
- Ouma JO, Marquez JG, Krafsur ES (2005) Macrogeographic population structure of the tsetse fly, *Glossina pallidipes* (Diptera : Glossinidae). B Entomol Res 95: 437–447.
- Bermond G, Ciosi M, Lombaert E, Blin A, Boriani M, et al. (2012) Secondary Contact and Admixture between Independently Invading Populations of the Western Corn Rootworm, *Diabrotica virgifera virgifera* in Europe. PLoS One 7: e50129.
- Ciosi M, Miller NJ, Toepfer S, Estoup A, Guillemaud T (2011) Stratified dispersal and increasing genetic variation during the invasion of Central Europe by the western corn rootworm, *Diabrotica virgifera virgifera*. Evol Appl 4: 54–70.
- Ciosi M, Toepfer S, Li H, Haye T, Kuhlmann U, et al. (2009) European populations of *Diabrotica virgifera virgifera* are resistant to aldrin, but not to methyl parathion. J Appl Entomol 133: 307–314.
- Lombaert E, Guillemaud T, Thomas CE, Handley LJL, Li J, et al. (2011) Inferring the origin of populations introduced from a genetically structured native range by approximate Bayesian computation: case study of the invasive ladybird *Harmonia axyridis*. Mol Ecol 20: 4654–4670.
- Beaumont MA, Zhang WY, Balding DJ (2002) Approximate Bayesian computation in population genetics. Genetics 162: 2025–2035.
- Pritchard JK, Seielstad MT, Perez-Lezaun A, Feldman MW (1999) Population growth of human Y chromosomes: A study of Y chromosome microsatellites. Mol Biol Evol 16: 1791–1798.
- Tavare S, Balding DJ, Griffiths RC, Donnelly P (1997) Inferring coalescence times from DNA sequence data. Genetics 145: 505–518.
- Lombaert E, Guillemaud T, Cornuet JM, Malausa T, Facon B, et al. (2010) Bridgehead Effect in the Worldwide Invasion of the Biocontrol Harlequin Ladybird. PLoS One 5: e9743.
- Miller N, Estoup A, Toepfer S, Bourguet D, Lapchin L, et al. (2005) Multiple transatlantic introductions of the western corn rootworm. Science 310: 992.
- Bertorelle G, Benazzo A, Mona S (2010) ABC as a flexible framework to estimate demography over space and time: some cons, many pros. Mol Ecol 19: 2609–2625.
- Chan YL, Anderson CNK, Hadly EA (2006) Bayesian estimation of the timing and severity of a population bottleneck from ancient DNA. Plos Genet 2: e59.
- Guillemaud T, Beaumont MA, Ciosi M, Cornuet JM, Estoup A (2010) Inferring introduction routes of invasive species using approximate Bayesian computation on microsatellite data. Heredity 104: 88–99.
- Ouma JO, Beadell JS, Hyseni C, Okedi LM, Krafsur ES, et al. (2011) Genetic diversity and population structure of *Glossina pallidipes* in Uganda and western Kenya. Parasites Vector 4: 122.
- Kariithi HM, Ahmadi M, Parker AG, Franz G, Ros VI, et al. (2013) Prevalence and genetic variation of salivary gland hypertrophy virus in wild populations of the tsetse fly *Glossina pallidipes* from southern and eastern Africa. J Invertebr Pathol 112 Suppl: S123–132.
- FAO (1982) Description and keys for the identification of *Glassina* species. In: Pollock JN, editor. Training manual for tsetse control personnel Volume 1. Rome: Food and agriculture organization of the united nations.

- Hyseni C, Beadell JS, Gomez Ocampo Z, Ouma JO, Okedi LM, et al. (2011) The *G.m. morsitans* (Diptera: Glossinidae) genome as a source of microsatellite markers for other tsetse fly (Glossina) species. Molecular Ecology Resources Primer Database, Available at: http://tomato.bio.trinity.edu/manuscripts/11-3/mer-10-0402.pdf [Accessed January 2014].
- Molecular Ecology Resources Primer Development C, Agata K, Alasaad S, Almeida-Val VMF, ÁLvarez-Dios JA, et al. (2011) Permanent Genetic Resources added to Molecular Ecology Resources Database 1 December 2010–31 January 2011. Mol Ecol Resources 11: 586–589.
- Baker MD, Krafsur ES (2001) Identification and properties of microsatellite markers in tsetse flies *Glossina morsitans* sensu lato (Diptera: Glossinidae). Mol Ecol Notes 1: 234–236.
- Ouma JO, Cummings MA, Jones KC, Krafsur ES (2003) Characterization of microsatellite markers in the tsetse fly, *Glossina pallidipes* (Diptera : Glossinidae). Mol Ecol Notes 3: 450–453.
- Ouma JO, Marquez JG, Krafsur ES (2006) New polymorphic microsatellites in Glossina pallidipes (Diptera : Glossinidae) and their cross-amplification in other tsetse fly taxa. Biochem Genet 44: 471–477.
- Holleley CE, Geerts PG (2009) Multiplex Manager 1.0: a cross-platform computer program that plans and optimizes multiplex PCR. Biotechniques 46: 511–517.
- 44. Salin F . AutoBin, Available at: http://www4.bordeaux-aquitaine.inra.fr/ biogeco/Media/Ressources/Logiciels/Autobin [Accessed January 2014].v 0.9 ed: INRA. pp. AutoBin is Excel Macro written in Microsoft Visual Basic (VBA). It automatically analyzes raw data generated from microsatellites genotyping software such as STRand (Toonen et al., 2001) or GeneMapper (Applied Biosystems, USA). It can deal with unlimited number of loci and samples, with no consideration of the type of SSRs motifs. AutoBin helps the user to bin his data with visual alerts and format automatically the data for downstream analysis.
- Nei M (1987) Molecular Evolutionary Genetics. New York: Columbia University Press.
- Piry S, Alapetite A, Cornuet JM, Paetkau D, Baudouin L, et al. (2004) GENECLASS2: A software for genetic assignment and first-generation migrant detection. J Hered 95: 536–539.
- Raymond M, Rousset F (1995) Genepop (version. 1.2), a population genetics software for exact tests and ecumenicism. J Hered 86: 248–249.
- Rousset F (2008) GENEPOP ' 007: a complete re-implementation of the GENEPOP software for Windows and Linux. Mol Ecol Resources 8: 103–106.
- Goudet J (2001) FSTAT, a program to estimate and test gene diversities and fixation indices (version 2.9.3). Updated from Goudet (1995).
- Raymond M, Rousset F (1995) An exact test for population differentiation. Evolution 49: 1280–1283.
- Benjamini Y, Hochberg Y (1995) Controlling the False Discovery Rate a Practical and Powerful Approach to Multiple Testing. J Roy Stat Soc B Met 57: 289–300.
- Weir BS, Cockerham C (1984) Estimating F-statistics for the analysis of population structure. Evolution 38: 1358–1370.
- Cornuet JM, Luikart G (1996) Description and power analysis of two tests for detecting recent population bottlenecks from allele frequency data. Genetics 144: 2001–2014.
- Piry S, Luikart G, Cornuet JM (1999) BOTTLENECK: A computer program for detecting recent reductions in the effective population size using allele frequency data. J Hered 90: 502–503.
- Estoup A, Wilson IJ, Sullivan C, Cornuet JM, Moritz C (2001) Inferring population history from microsatellite and enzyme data in serially introduced cane toads, *Bufo marinus*. Genetics 159: 1671–1687.
- Excoffier L, Estoup A, Cornuet JM (2005) Bayesian analysis of an admixture model with mutations and arbitrarily linked markers. Genetics 169: 1727–1738.
- Luikart G, Allendorf FW, Cornuet JM, Sherwin WB (1998) Distortion of allele frequency distributions provides a test for recent population bottlenecks. J Hered 89: 238–247.
- Pascual M, Chapuis MP, Mestres F, Balanya J, Huey RB, et al. (2007) Introduction history of *Drosophila subobscura* in the New World: a microsatellitebased survey using ABC methods. Mol Ecol 16: 3069–3083.
- Rannala B, Mountain JL (1997) Detecting immigration by using multilocus genotypes. Proc Natl Acad Sci U S A 94: 9197–9201.
- Ciosi M, Miller NJ, Kim KS, Giordano R, Estoup A, et al. (2008) Invasion of Europe by the western corn rootworm, *Diabrotica virgifera virgifera*: multiple transatlantic introductions with various reductions of genetic diversity. Mol Ecol 17: 3614–3627.
- Pritchard JK, Stephens M, Donnelly P (2000) Inference of population structure using multilocus genotype data. Genetics 155: 945–959.
- Coombs JA, Letcher BH, Nislow KH (2008) CREATE: a software to create input files from diploid genotypic data for 52 genetic software programs. Mol Ecol Resour 8: 578–580.
- Falush D, Stephens M, Pritchard JK (2003) Inference of population structure using multilocus genotype data: Linked loci and correlated allele frequencies. Genetics 164: 1567–1587.
- Evanno G, Regnaut S, Goudet J (2005) Detecting the number of clusters of individuals using the software STRUCTURE: a simulation study. Mol Ecol 14: 2611–2620.

- Earl DA, vonHoldt BM (2012) STRUCTURE HARVESTER: a website and program for visualizing STRUCTURE output and implementing the Evanno method. Conserv Genet Resour 4: 359–361.
- Cornuet JM, Ravigne V, Estoup A (2010) Inference on population history and model checking using DNA sequence and microsatellite data with the software DIYABC (v1.0). BMC Bioinformatics 11:401.
- Cornuet JM, Santos F, Beaumont MA, Robert CP, Marin JM, et al. (2008) Inferring population history with DIY ABC: a user-friendly approach to approximate Bayesian computation. Bioinformatics 24: 2713–2719.
- Garza JC, Williamson EG (2001) Detection of reduction in population size using data from microsatellite loci. Molecular Ecology 10: 305–318.
- Choisy M, Franck P, Cornuet JM (2004) Estimating admixture proportions with microsatellites: comparison of methods based on simulated data. Mol Ecol 13: 955–968.
- Magona JW, Okuna NM, Katabazi BK, Omollo P, Okoth JO, et al. (1998) Control of tsetse and animal trypanosomosis using a combination of tsetsetrapping, pour-on and chemotherapy along the Uganda-Kenya border. Rev Elev Med Vet Pay 51: 311–315.
- Hargrove JW, Langley PA (1993) A field trial of pyriproxyfen-treated targets as an alternative method for controlling tsetse (Diptera, Glossinidae). B Entomol Res 83: 361–368.
- Chakraborty R, Jin L (1993) A unified approach to study hypervariable polymorphisms: Statistical considerations of determining relatedness and population distances. In: Pena SDJ, Chakraborty R, Epplen JT, Jeffreys AJ, editors. DNA Fingerprinting: State of the Science: Birkhäuser Basel. pp. 153– 175.
- Goldstein DB, Linares AR, Cavallisforza LL, Feldman MW (1995) Genetic absolute dating based on microsatellites and the origin of modern humans. Proc Natl Acad Sci U S A 92: 6723–6727.
- Nesbitt SAT, Gooding RH, Rolseth BM (1990) Genetic variation in 2 field populations and a laboratory colony of *Glossina* pallidipes (Diptera, Glossinidae). J Med Entomol 27: 586–591.
- Hoffman JI, Grant SM, Forcada J, Phillips CD (2011) Bayesian inference of a historical bottleneck in a heavily exploited marine mammal. Mol Ecol 20: 3989– 4008.

- Girod C, Vitalis R, Leblois R, Freville H (2011) Inferring Population Decline and Expansion From Microsatellite Data: A Simulation-Based Evaluation of the Msvar Method. Genetics 188: 165–179.
- Hoban SM, Gaggiotti OE, Bertorelle G (2013) The number of markers and samples needed for detecting bottlenecks under realistic scenarios, with and without recovery: a simulation-based study. Mol Ecol 22: 3444–3450.
- International Atomic Energy Agency IAEA (2003) Automation for tsetse mass rearing for use in sterile insect technique programmes, Final report of a coordinated research project 1995–2001, Available at: http://www-pub.iaea.org/ MTCD/publications/PDF/te_1353_web.pdf [Accessed January 2014].
- van Etten J (1981) A comparison of the performance of laboratory colonies of Glossina pallidipes Austen from 2 allopatric populations in Kenya. Insect Sci Appl 1: 177–183.
- 80. Malele I (2013) Tsetse and Trypanosomiasis Research Institute, Tanzania, pers. com.
- Murilla G (2013) Kenya Agricultural Research Institute-Trypanosomiasis Research Centre, Kenya, pers. com.
- Abd-Alla AMM, Parker AG, Vreysen MJB, Bergoin M (2011) Tsetse Salivary Gland Hypertrophy Virus: Hope or Hindrance for Tsetse Control? Plos Neglect Trop D 5: e1220.
- Langley PA (1989) Laboratory colonization of the tsetse fly *Glossina pallidipes* Austen (Diptera, Glossinidae) using an *in vitro* feeding method. B Entomol Res 79: 429–435.
- Moloo SK (1992) A comparison of colony performance of *Glossina pallidipes* originating from 2 allopatric populations in Kenya. Med Vet Entomol 6: 399– 400.
- Ochieng RS, Otieno LH, Banda HK (1987) Performance of the tsetse fly *Glossina* pallidipts reared under simple laboratory conditions. Entomol Exp Appl 45: 265– 270.
- Facon B, Hufbauer RA, Tayeh A, Loiseau A, Lombaert E, et al. (2011) Inbreeding Depression Is Purged in the Invasive Insect *Harmonia axyridis*. Curr Biology 21: 424–427.
- Glémin S (2003) How Are Deleterious Mutations Purged? Drift versus Nonrandom Mating. Evolution 57: 2678–2687.